Author's response to reviews

Title: Management of Headache Disorders: Design of a Randomised Clinical Trial Screening for Prognostic Patient Characteristics

Authors:

Willem J De Hertogh (willem.de.hertogh@vub.ac.be)
Peter H Vaes (pvaes@vub.ac.be)
Dirk Devroey (ddevroey@vub.ac.be)
Steven Truijen (s.truijen@ha.be)
William Duquet (wduquet@vub.ac.be)
Rob A. B. Oostendorp (oostendorp@paramedisch.org)

Version: 4 Date: 26 April 2007

Author's response to reviews:

Dear Editor,

we submit a revision of our manuscript. The following changes have been made:

Title page - The E-mail addresses of all authors are provided

The NCT number from the title in both the manuscript and submission system has been removed.

The line numbers have been removed.

Competing interests - The formulation 'The authors declare that they have no competing interests' has been inserted.

Funding is now included as a subheading within acknowledgements. An additional acknowledgement has been added.

Websites are now cited from the main reference list.

Yours sincerely,

Willem De Hertogh